Author's response to reviews

Title: Successful reduction of hospital-acquired methicillin-resistant Staphylococcus aureus in a urology ward: a 10-year study

Authors:

Manabu Tatokoro (tatokoro.uro@tmd.ac.jp)
Kazunori Kihara (k-kihara.uro@tmd.ac.jp)
Hitoshi Masuda (hitoshi.masuda@jfcr.or.jp)
Masaya Ito (m-ito.uro@tmd.ac.jp)
Soichiro Yoshida (6030uro@tmd.ac.jp)
Toshiki Kijima (toshiki.kijima@gmail.com)
Minato Yokoyama (mntykym.uro@tmd.ac.jp)
Kazutaka Saito (kz-saito.uro@tmd.ac.jp)
Fumitaka Koga (f-koga.uro@tmd.ac.jp)
Satoru Kawakami (kawakami@saitama-med.ac.jp)
Yasuhisa Fujii (y-fujii.uro@tmd.ac.jp)

Version: 3 Date: 23 June 2013

Author's response to reviews: see over
Dr. Hayley Henderson  
Executive Editor, BMC Urology  

Re: MS: 4449202749214421 - Successful reduction of hospital-acquired methicillin-resistant Staphylococcus aureus in a urology ward: a 10-year study  

Dear Dr. Hayley Henderson,  

Thank you very much for giving us an opportunity to revise our original manuscript. The comments provided by reviewers were helpful and constructive. We have corrected our manuscript according to the points brought up by reviewers. We have outlined these points on separate sheets.  

Please find the revised manuscript that we hope you will find acceptable for publication in BMC Urology. Thank you in advance for considering our manuscript.  

Sincerely yours,  

Hitoshi Masuda, MD, PhD  
Department of Urology, Tokyo Medical and Dental University, Graduate School  
1-5-45, Yushima, Bunkyo-ku, Tokyo 113-8519, Japan  
E-mail: hi-masu.uro@tmd.ac.jp